A web-based self-management programme for people with type 2 diabetes: the HeLP-Diabetes research programme including RCT

Elizabeth Murray, 1* Jamie Ross, 1 Kingshuk Pal, 1 Jinshuo Li, 2 Charlotte Dack, 1 Fiona Stevenson, 1 Michael Sweeting, 3 Steve Parrott, 2 Maria Barnard, 4 Lucy Yardley, 5 Susan Michie, 6 Carl May, 7 David Patterson, 4 Ghadah Alkhaldi, 1,8 Brian Fisher, 9 Andrew Farmer 10 and Orla O'Donnell 1

- ¹Research Department of Primary Care and Population Health, University College London, London, UK
- ²Department of Health Sciences, University of York, Heslington, York, UK
- ³Cardiovascular Epidemiology Unit, Department of Public Health and Primary Care, University of Cambridge, Cambridge, UK
- ⁴Whittington Hospital, Whittington Health NHS Trust, London, UK
- ⁵Department of Psychology, University of Southampton, Southampton, UK
- ⁶Centre for Behaviour Change, Research Department of Clinical, Educational and Health Psychology, University College London, London, UK
- ⁷Faculty of Health Sciences, University of Southampton, Southampton, UK
- ⁸Community Health Sciences Department, College of Applied Medical Sciences, King Saud University, Riyadh, Saudi Arabia
- ⁹Patient Access to Electronic Records Systems Ltd (PAERS), Evergreen Life, Manchester, UK
- ¹⁰Nuffield Department of Primary Care Health Sciences, University of Oxford, Oxford, UK

Declared competing interests of authors: Elizabeth Murray reports grants from the National Institute for Health Research (NIHR) and National School of Primary Care Research during the conduct of the study, and grants from NIHR outside the submitted work. She is also managing director of the not-for-profit HeLP Digital community interest company (CIC), which was established to disseminate the HeLP-Diabetes programme across the NHS. She does not take any remuneration for this work. Kingshuk Pal reports personal fees from HeLP Digital CIC outside the submitted work. Fiona Stevenson reports grants from the NIHR during the conduct of this study. Maria Barnard reports sponsorship for attendance at educational conferences from Novo Nordisk A/S, personal fees from Janssen Pharmaceutica NV and that Janssen-Cilag International NV Sponsor was a sponsor of the multicentre CREDENCE trial (which Whittington Health NHS Health Trust is a site for) outside the submitted work. Lucy Yardley reports grants from NIHR during the conduct of the study; grants from NIHR, the Engineering and Physical Sciences Research Council, Medical Research Council (MRC), medical charities and European Commission outside the submitted work; and

^{*}Corresponding author elizabeth.murray@ucl.ac.uk

membership of the Health Technology Assessment Efficient Study Designs Board and Public Health Research Research Funding Board. Susan Michie reports grants during the conduct of the study from NIHR, the Engineering and Physical Sciences Research Council, MRC, the National Institute for Health and Care Excellence, the Department of Health and Social Care, Public Health England, Cancer Research UK, the British Psychological Society and EC outside the submitted work. David Patterson was one of the founders, and is now Chief Medical Officer, of Helicon Health Ltd, a spin-out company from UCL Business. He does not believe there is any actual conflict of interest between Helicon Health Ltd and this work – other than the fact that his knowledge of information technology and devices was growing during this academic work. Ghadah Alkhaldi reports personal fees from the Saudi Arabian Cultural Bureau, outside the submitted work. Brian Fisher reports that he is the Director of Patient Access to Electronic Record Systems Ltd (now Evergreen Life), which was intended to offer links from the HeLP-Diabetes programme to the patient record outside the submitted work. Orla O'Donnell reports working as the Chief Operating Officer for the HeLP Digital CIC from the end of the programme grant until 1 May 2017. Andrew Farmer reports grants from NIHR and MRC outside the submitted work.

Published September 2018 DOI: 10.3310/pgfar06050

Plain English summary

The HeLP-Diabetes research programme including RCT

Programme Grants for Applied Research 2018; Vol. 6: No. 5

DOI: 10.3310/pgfar06050

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain English summary

We developed, trialled and evaluated the HeLP-Diabetes programme, which is an online self-management programme for people with type 2 diabetes mellitus. To ensure that it met the needs of patients and health-care professionals (HCPs), we worked with patients, professionals, psychologists, sociologists, health services researchers, software engineers and web designers.

We took a broad approach to self-management and addressed the need for patients to adopt healthier behaviours, work with HCPs and manage negative emotions and changing feelings about themselves. Our goal was to help them to manage these different tasks throughout their illness.

We evaluated the HeLP-Diabetes programme in a randomised controlled trial. Volunteers were referred randomly to either the HeLP-Diabetes programme website or a simpler website, which was similar to NHS Choices. Although the people who volunteered for the trial had better control of their diabetes mellitus than most patients, after 12 months the people using the HeLP-Diabetes programme website had clinically better diabetes mellitus control than those using the simple website.

Overall, there was no difference in the levels of people's distress but the HeLP-Diabetes programme appeared to reduce distress among participants who had been more recently diagnosed with diabetes mellitus. The economic analysis showed that the HeLP-Diabetes programme improved outcomes and reduced costs.

The evaluation also showed that the programme was acceptable to HCPs and fitted well with NHS priorities. When HCPs invested time into helping people access the HeLP-Diabetes programme, it was possible to overcome the 'digital divide': over half of users who were helped by HCPs came from black or ethnic minority backgrounds, one-third had had no education after leaving school at 16 years old and one-third described themselves as having only 'basic' computer skills.

Programme Grants for Applied Research

ISSN 2050-4322 (Print)

ISSN 2050-4330 (Online)

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full PGfAR archive is freely available to view online at www.journalslibrary.nihr.ac.uk/pgfar. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

Criteria for inclusion in the Programme Grants for Applied Research journal

Reports are published in *Programme Grants for Applied Research* (PGfAR) if (1) they have resulted from work for the PGfAR programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Programme Grants for Applied Research programme

The Programme Grants for Applied Research (PGfAR) programme, part of the National Institute for Health Research (NIHR), was set up in 2006 to produce independent research findings that will have practical application for the benefit of patients and the NHS in the relatively near future. The Programme is managed by the NIHR Central Commissioning Facility (CCF) with strategic input from the Programme Director.

The programme is a national response mode funding scheme that aims to provide evidence to improve health outcomes in England through promotion of health, prevention of ill health, and optimal disease management (including safety and quality), with particular emphasis on conditions causing significant disease burden.

For more information about the PGfAR programme please visit the website: http://www.nihr.ac.uk/funding/programme-grants-for-applied-research.htm

This report

The research reported in this issue of the journal was funded by PGfAR as project number RP-PG-0609-10135. The contractual start date was in March 2011. The final report began editorial review in November 2016 and was accepted for publication in July 2017. As the funder, the PGfAR programme agreed the research questions and study designs in advance with the investigators. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The PGfAR editors and production house have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the final report document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, CCF, NETSCC, PGfAR or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the PGfAR programme or the Department of Health and Social Care.

© Queen's Printer and Controller of HMSO 2018. This work was produced by Murray et al. under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

NIHR Journals Library Editor-in-Chief

Professor Tom Walley Director, NIHR Evaluation, Trials and Studies and Director of the EME Programme, UK

NIHR Journals Library Editors

Professor Ken Stein Chair of HTA and EME Editorial Board and Professor of Public Health, University of Exeter Medical School, UK

Professor Andrée Le May Chair of NIHR Journals Library Editorial Group (HS&DR, PGfAR, PHR journals)

Dr Martin Ashton-Key Consultant in Public Health Medicine/Consultant Advisor, NETSCC, UK

Professor Matthias Beck Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Dr Peter Davidson Director of the NIHR Dissemination Centre, University of Southampton, UK

Ms Tara Lamont Scientific Advisor, NETSCC, UK

Dr Catriona McDaid Senior Research Fellow, York Trials Unit, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Professor of Wellbeing Research, University of Winchester, UK

Professor John Norrie Chair in Medical Statistics, University of Edinburgh, UK

Professor John Powell Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK

Professor James Raftery Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Great Ormond Street Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Jim Thornton Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Professor Martin Underwood Director, Warwick Clinical Trials Unit, Warwick Medical School, University of Warwick, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk